Author's response to reviews

Title: Remaining life expectancy among older people in a rural area of Vietnam: trends and socioeconomic inequalities during a period of multiple transitions

Authors:

Le V Hoi (hoilv@yahoo.com)
Ho D Phuc (hodang54@yahoo.com)
Truong V Dung (vietdzung52@yahoo.com)
Nguyen TK Chuc (ntkchuc@yahoo.com)
Lars Lindholm (lars.lindholm@epiph.umu.se)

Version: 4 Date: 9 November 2009

Author's response to reviews:

Dear the Editors,

Thank you very much for accepting, in principle, our manuscript, as well as providing very nice comments for further improving the manuscript from a reviewer before a publication. We have revised it accordingly with the comments.

Best regards,

LV Hoi